Author's response to reviews

Title: Does acute maternal stress in pregnancy affect infant health outcomes? Examination of a large cohort of infants born after the terrorist attacks of September 11, 2001

Authors:

Skye M. Endara (sendara@gmail.com)
Margaret A.K. Ryan (margaret.ryan@med.navy.mil)
Carter J. Sevick (carter.sevick@med.navy.mil)
Ava Marie S. Conlin (ava.conlin@med.navy.mil)
Caroline A. Macera (cmacera@mail.sdsu.edu)
Tyler C. Smith (tyler.c.smith@med.navy.mil)

Version: 3 Date: 19 June 2009

Author's response to reviews: see over
BMC Public Health  
BioMed Central Ltd  
Middlesex House  
34-42 Cleveland Street  
London W1T 4LB, UK

Dear BioMed Central Editorial Team,

Enclosed please find our revised manuscript “Does acute maternal stress in pregnancy affect infant health outcomes? Examination of a large cohort of infants born after the terrorist attacks of September 11, 2001.”

We are pleased that our previous revisions adequately addressed the reviewer comments, and we have incorporated referee #1’s suggestion into this final manuscript.

As corresponding author, please contact me if I can provide any additional information.

Respectfully,

Ava Marie S. Conlin, DO, MPH  
DoD Center for Deployment Health Research  
Naval Health Research Center, San Diego, CA, USA  
Phone: 619-767-4489  
Fax: 619-553-7601  
Email: ava.conlin@med.navy.mil